Author's response to reviews

Title: Massive Right Sided Bochdalek Hernia with two Unusual Findings: a case report.

Authors:

Subrato J Deb (sdeb@wmhs.com)

Version: 4 Date: 5 June 2011

Author's response to reviews:

Subrato J. Deb, MD, FCCP June 5, 2011
Cardiovascular and Thoracic Surgery
Western Maryland Regional Medical Center
Cumberland, Maryland 21502
Professor Michael Kidd
Editor in Chief
Journal of Medical Case Reports
Dear Professor Kidd,

Please accept this resubmission to your journal of the manuscript “Massive Right Sided Bochdalek Hernia with two Unusual Findings: a case report”.

Here is the point by point revisions to the reviewers most recent comments.

Dr. Vagefi’s comments:
1. In Figure 4, comment is added to the expanded legend that the right hemi-diaphragm is elevated due to the presence of prosthetic patch reconstruction.
2. A sentence has been added to the discussion portion of the manuscript stating the patient did not have any stigmata of end stage liver disease.
3. The CT images have been labeled.

Dr. Yamauchi did not have any further comments to the revised manuscript.

Thank you again for this consideration. I hope you will find this revised manuscript acceptable for publication to your journal.

Sincerely,

Subrato J. Deb, MD